ACAM

Evidence of COVID-19 Vaccine Safety: A Review of Reported Adverse Events

Salem Hussen Mosfer Alsaad (1), Ojeam Mana Saleh Al Mutared (2), Wael Hamad Al Jamhoor (3), Saleh Mohammed Hamad Al Yami (4), Khaled Hassan Hadi Al Hokash (5), Saleh Mohaimeed Saleh Al Sharmah (6), Masoud Saleh Ahmed Al Haydar (7), Yahya Ali Al qanass (1)

- (1) Nursing Technician at King Khalid Hospital in Najran, Saudi Arabia.
- (2) X-Ray Technician at King Khalid Hospital in Najran, Saudi Arabia.
- (3) X-Ray Technician at King Khalid Hospital, Najran, Saudi Arabia.
- (4) Emergency Medicine Ambulance at Mental Health Complex in Najran, Najran, Saudi Arabia.
- (5) Medical Laboratory Technician at King Khalid Hospital in Najran, Najran, Saudi Arabia.
- (6) Emergency Medical Services at King Khalid Hospital in Najran, Najran, Saudi Arabia.
- (7) Nursing Specialist at King Khalid Hospital Najran, Najran, Saudi Arabia.

Received 21/10/2022; revised 11/11/2022; accepted 14/12/2022

*Corresponding author

Abstract

Introduction: Vaccine hesitancy, influenced by complex behavioral factors such as the "theory of planned behavior," necessitates both quantitative and qualitative research to uncover the underlying causes of reluctance toward COVID-19 vaccination, which could undermine pandemic management efforts. This study focuses on assessing the reported side effects of COVID-19 vaccines as documented in scholarly article.

Methods: We targeted studies of a randomized controlled trial design to assess COVID-19 vaccine side effects. Databases including PubMed, Web of Science, Embase, CINAHL, PsycINFO, LILACS, SCOPUS, ClinicalTrials.gov, International Clinical Trials Registry Platform (ICTRP), medRxiv.org, biorxiv.org, preprints.org, and the Cochrane Library were searched up to November 2022. Three independent reviewers were involved in the study selection, methodological quality assessment, and data extraction processes. A meta-analysis was conducted where feasible, and findings were summarized using the Grading of Recommendations, Assessment, Development, and Evaluations framework.

Results: A variety of vaccine-related adverse effects were identified, including anaphylaxis, fever, musculoskeletal pain, headaches, fatigue, nausea, chills, skin redness, swollen lymph nodes, tingling sensations, vomiting, dizziness, heart rhythm anomalies, blood pressure fluctuations, and facial paralysis. The initial 15-day period post-vaccination was identified as carrying a heightened risk for these adverse effects. Our search yielded 29 results from PubMed and one from manual searching. To date, eight COVID-19 vaccine candidates have received emergency use authorization in various countries, and 12 are undergoing clinical trials tell the third quarter of this year.

Conclusions: Given the expedited emergency approval of COVID-19 vaccines without the full completion of clinical trials, it is imperative to rigorously establish their safety. The available information on these vaccines and their clinical trial safety outcomes remain to be fully determined.

Keywords: Vaccination, Adverse Reactions, Blood Clots, Safety Measures, COVID-19.

Introduction

The outbreak of severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2), known as COVID-19, escalated into a global pandemic, impacting virtually every nation [1]. As of July 2021, it has affected over 187 million individuals and resulted in more than four million fatalities [2]. This pandemic has triggered unprecedented economic challenges and social restrictions, posing a significant risk to healthcare systems worldwide. The World Health Organization has spearheaded international initiatives for both the primary and secondary prevention of COVID-19. Initial preventive measures against the pandemic included the adoption of infection control practices such as wearing face masks, maintaining hand hygiene, practicing social distancing, and shutting down communal spaces. Efforts such as active surveillance, prompt identification, isolation, and treatment of cases were implemented to mitigate the spread of the virus and avert the collapse of healthcare services.

Nevertheless, the development and distribution of a vaccine were identified as critical interventions, with numerous research institutions worldwide participating in vaccine development. By early 2021, several vaccines had successfully completed phase III trials and received authorization for widespread use, including the Pfizer-BioNTech vaccine (US -Germany), the Oxford University-AstraZeneca vaccine (UK), the Moderna vaccine (US), the Gamaleya-Sputnik V vaccine (Russia), Sinovac's COVID-19 vaccine (China), and the Johnson & Johnson vaccine (US). Despite these achievements, vaccine hesitancy emerged as a significant issue across various nations, fueled by misinformation related to the COVID-19 pandemic. The spread of conspiracy theories, particularly during the initial stages of the pandemic and amid quarantine measures, led to the activation of anti-vaccination movements [3]. These

misconceptions influenced public opinion about vaccines, contributing to widespread vaccine hesitancy, especially in light of reports concerning rare adverse effects associated with vaccination. Consequently, the availability of vaccines did not ensure their acceptance among the general populace or even healthcare professionals. Research indicates that public acceptance of vaccines plays a crucial role in achieving adequate vaccination coverage rates [4]. Various factors, including perceived risk of infection, educational level, ethnicity, and cultural beliefs, have been identified as influencing vaccine acceptance [5]. Addressing vaccine hesitancy necessitates an understanding of complex behavioral factors, such as the "theory of planned behavior" [6,7]. Both quantitative and qualitative research are essential for elucidating the factors contributing to reluctance towards COVID-19 vaccination, as this hesitancy could undermine efforts to control the pandemic. This study is dedicated to examining the thoroughly documented adverse effects of COVID-19 vaccination as reported in scientific literature.

Methods

We selected randomized controlled trial-type studies that evaluate the side effects of the COVID-19 vaccine. PubMed, Web of Science, Embase, CINAHL, PsycINFO, LILACS, SCOPUS, ClinicalTrials.gov, International Clinical Trials Registry Platform (ICTRP), medRxiv.org, biorxiv.org, preprints.org and the Cochrane Library were searched for eligible studies until November 2022. Three reviewers will independently screen and select studies, assess methodological quality and extract data. A metaanalysis were performed, if possible, and the Grading of Recommendations, Assessment, Development and Evaluations summary of findings were presented. We extracted and consolidate study-level data and extracted data from relevant studies. Collecting relevant study-level characteristics and experimental covariates were conducted. Select meta-analytic model. Evaluation quality of studies and estimation model parameters for complex relation-ships were done by 2 reviewers. The eligible studies were

Results and discussion

Different types of vaccine adverse effects have been described, such as anaphylaxis, fever, joint and muscular pain, headache, weakness, nausea, chills, erythema, lymphadenopathies, paresthesia, vomiting, dizziness, arrhythmias, changes in blood pressure, and facial paralysis [8]. During the first 15 days after vaccination, there is a high risk of developing all adverse symptoms. We found a total of 29 results from PubMed and one through manual searches. There are eight COVID-19 vaccine candidates have been approved for emergency use in different countries and 12 candidates in the clinical evaluation stage, which are developed using inactivated vaccine platform until August 2021. Remarkable progress has been made in the COVID-19 vaccines race while FDA provided full approval to the BNT162b2 vaccine on August 2021. T cells specific to SARS-CoV- 2 spike protein in the majority of vaccinated participants [9]. In this instance, the neutralizing antibody titers were obtained before the first dose, 21 days after the first dose, and seven and 14 days apart from the second dose, in which the antibody titers were highest at seven and 14 days apart from the second dose of vaccination. Moreover, the younger had higher antigen-binding IgG and virus-neutralizing antibody responses than the older [10]. BNT162b2 had an efficacy of 95% against COVID-19 after a two-dose (30 lg per dose and 21 days apart) vaccination [11]. The vaccine efficacy was generally identical across subgroups with individuals of different ages, sex, race, ethnicity, obesity, and preexisting conditions.

For this instance, the efficacy was 93.7-100% in participants (age of 16-55, [C 65, and C 75 years) who had no existing SARS-CoV-2 infection. Moreover, the efficacy also was similar in a separate analysis having participants with hypertension. Furthermore, the vaccine was 52.4% effective in preventing COVID-19 between two doses, but 90.5% effective within seven days interval of two-dose vaccination. A recent study from Public Health England (PHE) has shown that the vaccine had efficacy against Delta and Alpha variants

as 88% and 93%, respectively, after administration of the second dose [12]. Liu et al. reported that the vaccine had neutralizing activity against the Gamma variant, which is analogous to the action against the Alpha variant, and in the case of the Beta variant, the neutralizing activity was negligible.

On the contrary, the vaccine showed two dose-dependent systemic or local reactions, such as grade 1 and 2. A common adverse event was pain at the injection site, and other systematic events were also reported. The common systemic events were fatigue, chills, muscle pain, headache, and joint pain. Fever was also reported in some patients following the first and second dose vaccination but typically resolved within a day [13]. Gladly, no grade 4 adverse events were reported. However, grade 3 reactions such as sleep disturbance and pyrexia have been reported in a few participants.

The incidence rate was about 12.6 cases per million (second doses), which was less severe and usually resolved without medical intervention. In a casecontrol study conducted in Hong Kong, 16 out of 4,51,939 cases were confirmed with Bell's palsy/facial nerve palsy after the first dose, whereas another casecontrol study in Israel found no association between Bell's palsy and the vaccine. In the phase 1 trial, the vaccine induced both virus-neutralizing antibody and spike glycoprotein binding antibody responses in the participants aged 18-55 years [14]. The vaccine (both 25-lg and 100-lg doses) also elicited CD4 ? T-cell responses, predominantly Th1 responses stimulated by S-specific peptide pools with a fraction type 2 helper T-cell (Th2) responses. T-cell responses was also reported in the participants vaccinated with 100 lg dose (after the second dose).

The vaccine also had an efficacy of 93.6% in the SARS-CoV-2 seropositive participants . In the Qatar study, the vaccine showed 94.4% and 100% efficacy against the Alpha variant at C 28 days of the first dose and day 14 after the second vaccination dose, respectively. In the case of the Beta variant, the efficacy was 73.7% and 96.4% after 28 days of the first dose and 14 days of the second dose, respectively . Conversely, the most common adverse effect was pain at the injection site, and the others were tiredness,

headache, muscle ache, joint ache, chills, swollen lymph nodes in the vaccine injected arm, generally resolved within a day or two [15]. Further, nausea, vomiting, and fever were also reported. A significant portion of the participants has experienced these side effects after the second dose. Similar to the BNT162b2 vaccine, the incidence rate of adverse effects of the mRNA-1273 was about 12.6 cases per million dose (second dose) and had less severity and fatality, which did not require any medical intervention [16].

T-cell responses were detected. In addition, neutralizing antibody and spike protein-binding antibody responses were seen in most participants on day 29 after the single dose. The phase ENSEMBLE trial reported that the vaccine had an efficacy of 67% (single dose) against moderate to severe COVID-19 (symptomatic for at least 14 days) cases and 66% efficacy against patients having 28 days of onset [17]. The data from the phase ENSEMBLE trial also demonstrated that the vaccine efficacy was 85% against severe COVID-19 across all region studies and minimized hospitalization and death by COVID-19 after 28 days of vaccination. The effectiveness of the vaccine showed geographical variations such as in the USA, the efficacy was 72%, whereas, in South Africa (95% cases with the Beta variant), it was 64% against moderate to severe diseases and 81.7% against severe or critical illness after 28 days of vaccination. In Brazil (69% cases with the Gamma variant), the efficacy was 68.1% against moderate to severe diseases and 87.6% against severe or critical illnesses [18].

Conversely Sadoff et al. reported that the most common local reaction was injection site pain and the most frequent systemic reactions were headache, fatigue, myalgia, and nausea. The vaccine showed both humoral and cell-mediated immunity. The vaccine-induced IgG responses (28 days after vaccination) against the SARS-CoV-2 spike protein in all participants after two-dose vaccination were identical across the ages of 18-55, 56-69, and C 70 years. Moreover, neutralizing antibody responses have also been found after a booster dose (second dose) and were similar across all age groups. The vaccine also induced T-cell responses maximized at day 14 after vaccine administration (first dose). The interim

primary efficacy analysis by Ramasamy et al. revealed that the vaccine's efficacy was 70.4% after the second dose and 64.1% after the first dose in all participants . The USA Phase 3 trial by Astra-Zeneca revealed that the vaccine had 79% efficacy among the symptomatic COVID-19 and 100% efficacy in the severe and hospitalized COVID-19 patients. The efficacy was identical across diverse ethnicity and ages, and showed 80% efficacy in participants aged C 65 years [19].

Moreover, study data from Canada showed that the vaccine (Vaxzevria) had an efficacy of 90% against hospitalization or death caused by the Alpha variant, 87% against the Delta, and 82% against the Beta and Gamma variants after the first dose. Regarding symptomatic COVID-19 cases, the vaccine was 72% effective against the Alpha, 70% against the Delta, and 50% against the Beta and Gamma variants [20]. On the other hand, both local and systemic reactions were reported in all participants after vaccination. The most common local reaction was pain in the injection site, and other systematic reactions were fever, muscle ache, headache, chills, fatigue, malaise, and nausea, which were primarily mild and had 4-5 days of occurrence after vaccination [21].

The reactions were more common in younger than older adults (C 56 years). After the first dose, the vaccine also had a rare adverse event called thrombosis (blood clot) with thrombocytopenia syndrome. It could induce both humoral and cellular immune responses in all participants. The phase 3 clinical trial results revealed that it had induced virusneutralizing antibody response in the participants aged C 60 years. Furthermore, the vaccine efficacy was almost similar in the participants aged 18-60 years. Cell-mediated immune responses were also identified the participants where peripheral blood mononuclear cells had secreted IFN-c against SARS-CoV-2 spike glycoprotein (at day 28 after the first dose). The interim results of the phase 3 trial showed that the vaccine had an efficacy of 91.6% against COVID-19 (from day 21 after the first dose to the day of receiving the second dose). Although Logunov et al. found no serious adverse events, the most common local reaction was pain at the injection site, and other systematic reactions were headache, asthenia, and

muscle and joint pain. The vaccine-induced humoral responses were found as 100% seroconversion rate in 18-59 and C 60 years of participants [22]. In addition, in the 18-59 years group, the seroconversion rate was above 75% on day 14 after the first dose administration, and the rest of the participants were seroconverted on day 28. The 18-59 years group also had higher neutralizing antibody titers than the [60 years group. Sinopharm reported that the vaccine had an efficacy of 79.34% in the phase 1/2 trial, and in the phase 3 trial (multi-countries), the efficacy was 79% in both symptomatic and hospitalized patients after 14 days of vaccination.

The local reactions were pain at the vaccination site, flush, swelling, scleroma, rash, and itching. The systemic reactions were headache, fever, fatigue, muscle pain, joint ache, cough, breathing difficulty, nausea, diarrhea, and skin itching [23]. In the phase 1 trial, the vaccine (two doses) induced humoral responses against SARS-CoV-2 at 104 days after vaccination (3 months after the second dose). At the same time, serum neutralizing antibodies were also detected among all the participants. In the phase 2 trial, the vaccine (after two doses) induced prominent Th1 immune responses with a high level of IFN-c. The immune responses were significantly higher in the phase 2 trial than in the phase 1 trial. Moreover, the vaccine could induce both memory B-cell and T-cell responses, where the latter expressed the memory phenotype marker CD45RO?.

Bharat Biotech claimed that the vaccine had an efficacy of 77.8% against symptomatic, 63.6% against asymptomatic, and 93.4% against severe symptomatic COVID-19 cases. Contrarily, the most common local adverse reaction was pain at the injection site, followed by other systematic reactions such as headache, fatigue, vomiting, and fever. The reactions were mild to moderate and more frequent after the first dose. However, one severe adverse reaction of viral pneumonitis was reported in the phase 1 trial [24]. The vaccine had been induced both humoral and cellular responses in most of the participants. The vaccine elicited RBD (receptor binding domain) binding antibodies (four-fold higher) and live virus binding antibodies (four-fold higher) in 94-100% and 50-75% of the participants. Moreover T-cell responses were

detected on day 14 after vaccination and antibodies on day 28, and both were in peaked value at the times mentioned above. T cells was reported in all the participants. However, when pre-existing anti-Ad5 immunity was high, both the antibody and T-cell response were slightly reduced.

The Ad5-nCoV vaccine efficacy was 68.83% (single dose) against all symptomatic and 95.47% against severe COVID-19 cases 14 days after vaccination. After 28 days of vaccination, the efficacy was 65.28% against all symptomatic and 90.07% against severe COVID-19 cases [25]. Conversely, the vaccine was reported with some common adverse reactions such as fever, fatigue, headache, and muscle ache. Those reactions were mild, moderate, and consistent among all the participants. Further, no severe adverse reactions were noted as a matter of concern. Until August 2021, the phase 3 trial result of the vaccine has not been published yet. In addition, the vaccine had a similar immune response in adults aged C 60 years [26]. The neutralizing antibody responses were reported in the older (C 60 years) and younger adults (18-59 years) with similar magnitude after two doses of vaccination (3 lg and 6 lg).

The study also reported a strong correlation between the neutralizing antibodies and the anti-RBD IgG antibodies. Moreover, weak T-cell responses were found in 18-59 years aged adults. In Brazil, the phase 3 trial demonstrated that the vaccine efficacy was 51% against symptomatic and 100% against severe SARS-CoV-2 infection after the second dose. In Turkey, the phase 3 trial revealed that the vaccine efficacy was 83.5% on 14 days after administration of the second dose [23]. Conversely, the most common adverse reactions were mild pain at the injection site, elevated blood pressure, headache, fatigue, dizziness, and rash. There were no grade 4 adverse events or fatality has been seen . However, the case-control study from Hong Kong revealed that the vaccine could cause Bell's palsy, where 28 out of 5,37,205 cases were confirmed with this type of adverse event [22. It induced humoral responses against SARS-CoV-2 in all participants. In the phase 2 trial, the virusneutralizing antibodies were found 14 days after the second dose with a seroconversion rate of 76% in people who received 25 lg of vaccine and 72% in

people who received 50 lg of the vaccine. However, the third dose of the vaccine was found with an increased seroconversion rate in both 25 lg and 50 lg dose groups estimating 97% and 93%, respectively. The most common adverse reactions were mild and transient pain at the injection site, redness, and swelling, resolved within 3-4 days after vaccination. The trials also documented that the neutralizing antibody responses were higher between 21 and 28 days after the second dose and highest after the third or booster dose administration [27]. Sinopharm reported that the vaccine had an efficacy of 72.51%. A recent study showed that the vaccine efficacy was 79% in both symptomatic and hospitalized patients after 14 days of vaccination and 78.1% against only symptomatic adult patients.

On the contrary, the most common adverse reaction was transient mild pain at the injection site. The second dose of the vaccine reported no adverse events in the participants after vaccination [28]. In the phase 2 trial of the candidate vaccine, ten times more neutralizing antibodies were found in the vaccinated people than in the recovered COVID-19 patient. Lowincome countries are not getting the spotlight of the vaccination program as only 0.3% of the total vaccines are administered in their arms. On the other hand, 83% of the total shots are applied in the high and upper-and middle-income countries to make their people vaccinated. The people of Africa got the lowest doses of vaccines, only 5.8 people per hundred population, whereas 88 people per hundred population brought under vaccination in North America. Furthermore, 91%, 68%, 64%, and 41% of people become vaccinated in Europe South America Asia, and Oceania, respectively (Fig. 3). Most of the wealthy countries pre-ordered vaccine doses, but the lowincome countries could not ensure enough amounts, such as the countries of Africa. As of August 2021, 87% of the total 4.62 billion doses of vaccines are obtained by high-income countries, while low-income countries just received 0.3 percent. Discrimination is clear as almost one of every four people of rich countries gets vaccinated, while it is one in over 500 in the low-income countries. There was a hope of herd immunity as different studies described that almost 60-70% of the population gained immunity through vaccinations or past exposure to the virus [29].

Conclusions

At present, it is very crucial to establish the safety of the COVID-19 vaccines when emergency approval is being granted to these vaccines without completion of all phases of clinical trials. Therefore, the current evidence reflected the safety generated from the results of published clinical trials of these vaccines. The details of these vaccines and clinical trials reporting the safety of COVID-19 vaccines are not conclusive. Despite all the vaccines' benefits and effectiveness, as mentioned previously, mild and negligible side effects have been observed. It can affect some plans in the screening or follow-up of cancerous patients simultaneously.

Conflict of interests

The authors declared no conflict of interests.

References

- 1. Oldenburg, Johannes., Klamroth, Robert., Langer, Florian., Albisetti, Manuela., Von Auer, Charis., Ay, Cihan., Korte, Wolfgang., Scharf, Rüdiger., Pötzsch, Bernd. and Greinacher, Andreas. Diagnosis and Management of Vaccine-Related Thrombosis following AstraZeneca COVID-19 Vaccination: Guidance Statement from the GTH. (2021) Hamostaseologie. 41(03); 184-189.
- 2. Soiza, Roy., Scicluna, Chiara., Thomson, Emma. and Ward, . Efficacy and safety of COVID-19 vaccines in older people. (2021) 50(2); 279-283.
- 3. Doroftei, Bogdan., Ciobica, Alin., Ilie, Ovidiu-Dumitru., Maftei, Radu., Ilea, Ciprian., Cheng, Chao-Min., Vashist, Sandeep. and De Mendoza, Carmen. Mini-Review Discussing the Reliability and Efficiency of COVID-19 Vaccines. (2021) Diagnostics. 11(4); 579.
- 4. Halim, Michael., Halim, Alice. and Tjhin, Yovita. COVID-19 Vaccination Efficacy and Safety Literature Review. (2021) J Immuno Allerg.
- 5. Manohar, Prasanth., Li, Brian., Wong, Han., De Albuquerque, Matilde., Machado, Veloso., Roberts, Bjelle., Van Kessel, Robin. and Mossialos, Elias. The Relationship Between the COVID-19

- Pandemic and Vaccine Hesitancy: A Scoping Review of Literature Until August 2021. (2021) Front. Public Health. 9;
- 6. Siddique, Soma. and Ahmed, Shaheer. COVID-19 Vaccines in Pakistan: Efficacy, Adverse Effects and Availability. (2021) J Islamabad Med Dental Coll. 10(2); 125-130.
- 7. Lu, Lu., Xiong, Weixi., Mu, Jie., Zhang, I., Zhang, I., Zou, Ling., Li, Weimin., Li, I., Sander, Josemir. and Zhou, Dong. The potential neurological effect of the COVID-19 vaccines: A review. (2021) Acta Neurol Scand. 144(1); 3-12.
- 8. Kaur, Simran. and Gupta, Vandana. COVID-19 Vaccine: A comprehensive status report. (2020) Virus Research. 288; 198114.
- 9. Chen, Yue., Xu, Zhiwei., Wang, Peng., Li, Xiao-Mei., Shuai, Zong-Wen., Ye, Dong-Qing. and Pan, Hai-Feng. New-onset autoimmune phenomena post-COVID-19 vaccination. (2022) Immunology. 165(4); 386-401.
- 10. Sessa, Francesco., Salerno, Moni., Esposito, Massimiliano., Nunno, Nunzio., Zamboni, Paolo., Pomara, Cristoforo. and Russo, Giorgio. Autopsy Findings and Causality Relationship between Death and COVID-19 Vaccination: A Systematic Review. (2021) JCM. 10(24); 5876.
- 11. Keshavarz, Pedram., Yazdanpanah, Fereshteh., Rafiee, Faranak. and Mizandari, Malkhaz. Lymphadenopathy Following COVID-19 Vaccination: Imaging Findings Review. (2021) Academic Radiology. 28(8); 1058-1071.
- 12. Brazete, C., Aguiar, A. and U M M A R Y, S. Thrombotic events and COVID-19 vaccines. (2021) int j tuberc lung dis. 25(9); 701-707.
- 13. Bilotta, Clio., Perrone, Giulio., Adelfio, Valeria., Spatola, Giovanni., Uzzo, Maria., Argo, Antonina., Zerbo, Stefania. and Tsang, Kwong. COVID-19 Vaccine-Related Thrombosis: A Systematic Review and Exploratory Analysis. (2021) Front. Immunol. 12;
- 14. Jeet Kaur, Rimple., Dutta, Siddhartha., Bhardwaj, Pankaj., Charan, Jaykaran., Dhingra, Sameer., Mitra, Prasenjit., Singh, Kavita., Yadav, Dharmveer., Sharma, Praveen. and Misra, Sanjeev. Adverse Events Reported From COVID-19 Vaccine Trials: A Systematic Review. (2021) Ind J Clin Biochem. 36(4); 427-439.

- 15. Wu, Qianhui., Dudley, Matthew., Chen, Xinghui., Bai, Xufang., Dong, Kaige., Zhuang, Tingyu., Salmon, Daniel. and Yu, Hongjie. Evaluation of the safety profile of COVID-19 vaccines: a rapid review. (2021) BMC Med. 19(1);
- 16. Rahman, Md., Habib, Md., Masum, Ullah., Shah, •. and Talukder, Asma. A comprehensive review on COVID-19 vaccines: development, effectiveness, adverse effects, distribution and challenges. (2022) VirusDis. 33(1); 1-22.
- 17. Ooi, Eng., Dhar, Arti., Petruschke, Richard., Locht, Camille., Buchy, Philippe., Guek, Jenny. and Low, Hong. Use of analgesics/antipyretics in the management of symptoms associated with COVID-19 vaccination. (2022) npj Vaccines. 7(1);
- 18. Pormohammad, Ali., Zare, Mohammad., Ghorbani, Saied., Mohamma, Mehdi., Razizadeh, Mohammad., Turner, Diana., Turner, Raymond., Rajendran, Chandra. and Jaganathan, K. Efficacy and Safety of COVID-19 Vaccines: A Systematic Review and Meta-Analysis of Randomized Clinical Trials. (2021) Vaccines. 9(5); 467.
- 19. Sadoff, J., Gray, G., Vandebosch, A., Cárdenas, V., Shukarev, G., Grinsztejn, B., ... & Douoguih, M. (2021). Safety and efficacy of singledose Ad26. COV2. S vaccine against Covid-19. New England Journal of Medicine, 384(23), 2187-2201.
- 20. Anand, P., & Stahel, V. P. (2021). The safety of Covid-19 mRNA vaccines: A review. Patient safety in surgery, 15(1), 1-9.
- 21. Gee, J., Marquez, P., Su, J., Calvert, G. M., Liu, R., Myers, T., ... & Shimabukuro, T. (2021). First month of COVID-19 vaccine safety monitoring—United States, December 14, 2020—January 13, 2021. Morbidity and mortality weekly report, 70(8), 283.
- 22. Anand, P., & Stahel, V. P. (2021). The safety of Covid-19 mRNA vaccines: A review. Patient safety in surgery, 15(1), 1-9.
- 23. Al Khames Aga, Q. A., Alkhaffaf, W. H., Hatem, T. H., Nassir, K. F., Batineh, Y., Dahham, A. T., ... & Traqchi, M. (2021). Safety of COVID-19 vaccines. Journal of medical virology, 93(12), 6588-6594.
- 24. Kaplan, R. M., & Milstein, A. (2021). Influence of a COVID-19 vaccine's effectiveness and safety profile on vaccination acceptance. Proceedings of the National Academy of Sciences, 118(10), e2021726118.

- 25. Ghiasi, N., Valizadeh, R., Arabsorkhi, M., Hoseyni, T. S., Esfandiari, K., Sadighpour, T., & Jahantigh, H. R. (2021). Efficacy and side effects of Sputnik V, Sinopharm and AstraZeneca vaccines to stop COVID-19; a review and discussion. Immunopathologia Persa, 7(2), e31-e31.
- 26. Singh, A., Khillan, R., Mishra, Y., & Khurana, S. (2022). The safety profile of COVID-19 vaccinations in the United States. American journal of infection control, 50(1), 15-19.
- 27. Saeed, B. Q., Al-Shahrabi, R., Alhaj, S. S., Alkokhardi, Z. M., & Adrees, A. O. (2021). Side effects and perceptions following Sinopharm COVID-19 vaccination. International Journal of Infectious Diseases, 111, 219-226.
- 28. Bogdanov, G., Bogdanov, I., Kazandjieva, J., & Tsankov, N. (2021). Cutaneous adverse effects of the available COVID-19 vaccines. Clinics in dermatology, 39(3), 523-531.
- 29. Hatmal, M. M. M., Al-Hatamleh, M. A., Olaimat, A. N., Hatmal, M., Alhaj-Qasem, D. M., Olaimat, T. M., & Mohamud, R. (2021). Side effects and perceptions following COVID-19 vaccination in Jordan: a randomized, cross-sectional study implementing machine learning for predicting severity of side effects. Vaccines, 9(6), 556.

